Author’s response to reviews

Title: Structured reporting of head and neck ultrasound examinations

Authors:

Benjamin Ernst (benjamin.ernst@unimedizin-mainz.de)
Mohamed Hodeib (Mohamed.Hodeib@hotmail.de)
Sebastian Strieth (sebastian.strieth@unimedizin-mainz.de)
Julian Künzel (julian.kuenzel@unimedizin-mainz.de)
Fabian Bischof (bischof.fabian@outlook.de)
Berit Hackenberg (berit.hackenberg@unimedizin-mainz.de)
Tilmann Huppertz (tilmann.huppertz@unimedizin-mainz.de)
Veronika Weber (veronika.weber@unimedizin-mainz.de)
Katharina Bahr (katharina.bahr@unimedizin-mainz.de)
Jonas Eckrich (jonas.eckrich@unimedizin-mainz.de)
Jan Hagemann (jan.hagemann@unimedizin-mainz.de)
Matthias Engelbarts (matthias.engelbarts@unimedizin-mainz.de)
Matthias Froelich (matthias.froelich@med.uni-muenchen.de)
Philipp Solbach (solbach.philipp@mh-hannover.de)
Richard Linke (richard.linke@kgu.de)
Christoph Matthias (christoph.matthias@unimedizin-mainz.de)
Wieland Sommer (sommer@mail.harvard.edu)
Sven Becker (sven.becker@unimedizin-mainz.de)

Version: 1 Date: 23 Jan 2019

Author’s response to reviews:
Dear ladies and gentlemen,

we revised our manuscript BMIM-D-18-00268 according to the reviewer’s suggestions and feel confident that the manuscript meets your expectations by now. Since the reviewer pointed out some language issues, the manuscript was checked by a professional medical language editing service and a thorough revision was carried out accordingly.

A point by point response letter explaining all changes within the manuscript as well as a manuscript version with track changes was uploaded as supplementary.

Thank you very much in advance for re-reviewing our manuscript!

Best regards,

Dr. Benjamin Ernst